\magnification=\magstep1
\nopagenumbers
\parindent=0pt
{\bf Math 225: Calculus III \hfill Spring 1997}


\centerline{\bf Course Information}


\medskip
TEXT:  Finney \& Thomas, Calculus, 2nd Edition, and Snow, Lecture Notes

\medskip
EXAM SCHEDULE:  There will be three one-hour exams and a two hour
final on the following dates:
$$\vbox{\halign{#\ \hfil&#\hfil\cr
Exam I &Tuesday, February 11, 8:00 A.M.\cr
Exam II &Thursday, March 20, 8:00 A.M.\cr
Exam III &Thursday, April 17, 8:00 A.M.\cr
Final &Thursday May 8, 1:45 P.M.\cr
}}$$
The exam locations will be announced. Exams may be made up only
with an excused absence from the Associate Vice President for
Resident Life. Conflicts with exams in other courses must be resolved
during the first week of classes.

\medskip
QUIZZES/ASIGNMENTS: Quizzes will be taken and assignments handed in at
your tutorials on the following dates:
$$\vbox{\halign{#\hfil\cr
Quiz 1 \& Assignment 1: Jan.~21/23\cr
Quiz 2 \& Assignment 2: Jan.~28/30\cr
Quiz 3 \& Assignment 3: Feb.~4/6\cr
Quiz 4 \& Assignment 4: Feb.~18/20\cr
Quiz 5 \& Assignment 5: Feb.~25/27\cr
Quiz 6 \& Assignment 6: Mar.~4/6\cr
Quiz 7 \& Assignment 7: Mar.~25/27\cr
Quiz 8 \& Assignment 8: Apr.~1/3\cr
Quiz 9 \& Assignment 9: Apr.~8/10\cr
Quiz 10 \& Assignment 10: Apr.~22/24\cr
}}$$
You may only attend the tutorial assigned to you. The assignments can
be found in the course directory and must be prepared using {\sl
Mathematica}. Each quiz is worth 5 points and each assignment is worth
10 points.

\medskip
COMPUTERS: You will be expected to work out assignments using {\sl
Mathematica} on the Macintoshes in CCMB 210 or any other lab that
has the appropriate software. Your grades will be sent to you
regularly through your campus e-mail account. You can obtain your
password at the Information Resource Center, 1st floor CCMB.

\medskip
OFFICE HOURS: Office hours will be posted in the Mathematics Building:
$$\vbox{\halign{#\ \hfil&#\hfil\ &#\hfil\cr
D. Snow &CCMB 203\cr
M. Anton &CCMB 336\cr
L. Feher &CCMB 220\cr
}}$$

\medskip
GRADES:  Grades will be based on a total of 600 points broken down as follows:
$$\vbox{\halign{#\ \hfil&#\ \hfil&#\hfil\cr
Exam I--III &3 @ 100  &= 300\cr
Final &&= 150\cr
Quizzes &10 @ 5 &= 50\cr
Assignments &10 @ 10 &= 100\cr
}}$$

\bye
